Author's response to reviews

Title: Movement control exercise versus general exercise to reduce disability in patients with low back pain and movement control impairment. A randomised controlled trial

Authors:

Jeannette Saner (seat@zhaw.ch)
Jan Kool (jan.kool@zhaw.ch)
Rob A de Bie (ra.debie@epid.unimaas.nl)
Judith M Sieben (j.sieben@maastrichtuniversity.nl)
Hannu Luomajoki (hannu.luomajoki@zhaw.ch)

Version: 2 Date: 18 August 2011

Author's response to reviews:

8401 Winterthur, 18. August 2011

Unser Zeichen: Seat

2. Coverletter Study protocol – revision - 1809559709573274

Movement control exercise versus general exercise to reduce disability in patients with low back pain and movement control impairment. A randomised controlled trial

Saner J, Kool J, de Bie RA, Sieben JM, Luomajoki H

Jeannette Saner

MSc
Dozent

jeannette.saner@zhaw.ch

Büro

Technikumstrasse 71
Postfach
CH- 8401 Winterthur

Tel. +41 58 934 63 70
Dear Sir,

The following revisions in our publication are following your recommendations of July 15, 2011

We have added the trial number to the abstract section

We have made minor changes to “authors contribution” and have added the funding detail in the “acknowledgement” section.

We are providing prove of ethical processes and national funding details by email to BMC Series Editorial.

Yours sincerely,

Zürcher Hochschule für Angewandte Wissenschaften

Jeannette Saner